Author's response to reviews

Title: Enhanced Relapse Prevention for Bipolar Disorder - ERP trial. A cluster randomised controlled trial to assess the feasibility of training care coordinators to offer enhanced relapse prevention for bipolar disorder.

Authors:

Fiona Lobban (fiona.lobban@manchester.ac.uk)
Carol Gamble (c.gamble@liverpool.ac.uk)
Peter Kinderman (p.kinderman@liverpool.ac.uk)
Lee Taylor (lee.taylor@penninecare.nhs.uk)
Claire Chandler (claire.chandler@liverpool.ac.uk)
Elizabeth Tyler (e.tyler@liverpool.ac.uk)
Sarah Peters (s.peters@liverpool.ac.uk)
Eleanor Pontin (e.pontin@liverpool.ac.uk)
William Sellwood (sellwood@liverpool.ac.uk)
Richard K Morriss (richard.morriss@nottingham.ac.uk)

Version: 2 Date: 28 January 2007

Author's response to reviews:

Competing Interests

The authors declare they have no competing interests.

Authors' contributions

FL was involved in the design, oversaw the implementation, produced initial draft of manual, and drafted the manuscript. CG provided statistical expertise in design, data collection and the manuscript. PK contributed to the design and supervision of data collection. LT developed the manual and a training package, provided the clinical training and supervision and collected data linked to this. CC managed and carried out recruitment and data collection. ET carried out recruitment and data collection and contributed to a draft of the manuscript. SP and EP jointly designed and managed the qualitative arm of the study. WS contributed to the training package and supervision of data collection. RM was involved in the design of the study, oversaw implementation of recruitment and clinical assessments, designed the training materials, and contributed to drafting the manuscript.